Author's response to reviews

Title: Induction of cytotoxic T lymphocytes primed with Tumor RNA-loaded Dendritic Cells in esophageal squamous cell carcinoma: preliminary step for DC vaccine design

Authors:

Mehran Gholamin (mgholamin@yahoo.com)
Omeed Moaven (omeedmoaven@gmail.com)
Moein Farshchian (moeinfarshchy@yahoo.com)
Mahmoud Mahmoudi (mahmoudim@mums.ac.ir)
Mojtaba Sankian (sankianm@mums.ac.ir)
Bahram Memar (memarb@mums.ac.ir)
Mohammad Naser Forghani (dr_forghani@yahoo.com)
Reza Malekzadeh (malekzadeh2@gmail.com)
Mohammad Taghi Rajabi-Mashhadi (rajabimt@mums.ac.ir)
Mohammad Reza Abbaszadegan (abbaszadeganmr@mums.ac.ir)

Version: 8 Date: 30 May 2010

Author's response to reviews: see over
Dear Editor,

I am resubmitting the manuscript titled “Induction of cytotoxic T lymphocytes primed with Tumor RNA-loaded Dendritic Cells in Esophageal Squamous Cell Carcinoma: preliminary step for DC vaccine design”. Minor typographic revisions are performed and the manuscript is ready for publication.

Sincerely yours,

Mohammad R. Abbaszadegan, MT(ASCP), Ph.D.
Associate Professor
Director, Division of Human Genetics
Avicenna Research Institute
Mashhad University of Medical Sciences
Mashhad, Iran postal code: 9196773117
Phone: +98511-7112611-5
Fax: +98511-7112616
abbaszadeganmr@mums.ac.ir